Author’s response to reviews

Title: The effects of different shapes of capsulorrhexis on postoperative refractive outcomes and the effective position of the intraocular lens in cataract surgery

Authors:

li shixu (751779333@qq.com)
Yiping Hu (409359997@qq.com)
Ran Guo (471749905@qq.com)
Yushuang Shao (1025039950@qq.com)
Jiangyue Zhao (zhaojiangyue@hotmail.com)
Jinsong Zhang (cmu_zhangjinsong1@163.com)
Jing Wang (wangjing@cmu.edu.cn)

Version: 2 Date: 07 Feb 2019

Author’s response to reviews:

Dear Editor:

Thank you for your kind instructive comments and suggestions.

We have revised most of the suggestions, and the forms have been uploaded as separate files. We are hoping to get your approval!

Thank you very much for consideration!

Sincerely Yours,

Dr. Shixu Li